Author's response to reviews

Title: Adenocarcinoma of the bladder following nephrogenic adenoma: a case report

Authors:

  Edwin Hungerhuber (Hungerhuber@gmx.de)
  Ekkehard Bach (Dr.Ekkehard.Bach@t-online.de)
  Arndt Hartmann (arndt.hartmann@klinik.uni-regensburg.de)
  Dominic Frimberger (frimberger@hotmail.com)
  Christian G Stief (Christian.Stief@uro.med.uni-muenchen.de)
  Dirk Zaak (Dirk.Zaak@uro.med.uni-muenchen.de)

Version: 4 Date: 11 April 2008

Author's response to reviews: see over
To the Editorial board

Journal of
Medical Case Reports

München, den 11/04/2008

Dear Editorial Board,

herewith we send our revised manuscript for a case report with the title:

Adenocarcinoma of the bladder following nephrogenic adenoma: a Case report.

We agree with all the amendments made by the editorial board.

The highlighted sections were completed:

1. The number {243} represents a reference which has been added (number 17)
2. DF meade the language corrections
3. The complete list of authors is provided (reference number 9)

Thank you very much in advance!

Sincerely yours

Edwin Hungerhuber